Author’s response to reviews

Title: Multifocal multiorgan ischaemia and infarction in a preterm infant due to intravenous cocaine use: A Case Report

Authors:

Ben C Reynolds (pinkdoc@doctors.org.uk)
Dawn MK Penman (dawn.penman@yorkhill.scot.nhs.uk)
Allan G Howatson (allan.howatson@yorkhill.scot.nhs.uk)
Lesley A Jackson (lesley.jackson@northglasgow.scot.nhs.uk)
Charles H Skeoch (Charles.skeoch@northglasgow.scot.nhs.uk)

Version: 3 Date: 29 April 2009

Author’s response to reviews: see over
To The Editor,

Many thanks for considering the re-submission of the article: Multifocal multigorgan ischaemia and infarction in a preterm infant due to intravenous cocaine use: A case report

Further to your reviewers comments:

Reviewer 1 made no suggestions re alteration

Reviewer 2:

Glucocorticoid has been changed to Betamethasone, and hypoprothrombinaemia to coagulopathy for consistency within the manuscript.
Consideration of haemolysis potentially causing the 'pink' urine is added
'Midline shift’ has been substituted instead of mass shift.
The sentence regarding the reorientation of care has been reworded
The mechanism of cocaine causing vasoconstriction has been added to the discussion
The polypharmacy sentence has been restructured and split into 2 sentences.
The conclusion has been altered to include coagulation screening
Table One has been shortened
Grammar has been checked throughout the manuscript

Many thanks,

Ben Reynolds